Author's response to reviews

Title: How effective is tetracaine 4% gel, before a peripherally inserted central catheter, in reducing procedural pain in infants: a randomized double-blind placebo controlled trial [ISRCTN75884221]

Authors:

Brigitte Lemyre (blemyre@ottawahospital.on.ca)
Rebecca Sherlock (rebeccalynsherlock@hotmail.com)
Debora Hogan (Hogan@cheo.on.ca)
Isabelle Gaboury (gaboury@cheo.on.ca)
Colline Blanchard (blanchard_c@cheo.on.ca)
David Moher (dmoher@uottawa.ca)

Version: 6 Date: 24 April 2006

Author's response to reviews: see over
To: Biomed Central Editorial Board

Ottawa, April 24th, 2006

RE: formatting changes requested prior to publishing on BMC Medicine of our manuscript  How effective is Amethocaine 4% gel, before a peripherally inserted central catheter, in reducing procedural pain in infants: a randomized double-blind placebo controlled trial [ISRCTN75884221]

Dear editor,

All suggested editing changes have been accepted. The query about reference #28 has been addressed. I have checked how this systematic review is referenced in Pubmed and have made the modification in the document.

I hope the manuscript is now acceptable and ready for publication. Thank you again for this opportunity.

Brigitte Lemyre MD FRCPC